Author's response to reviews

Title: Association of HbA1c and cardiovascular and renal disease in an adult Mediterranean population

Authors:

Domingo Hernandez (domingohernandez@gmail.com)
Ana Espejo-Gil (ana_eg_@hotmail.com)
M Rosa Bernal-Lopez (robelopaijiu@yahoo.es)
Jose Mancera-Romero (jmancerar@papps.org)
Antonio J Baca-Osorio (abalo57@hotmail.com)
Francisco J Tinahones (fjtinahones@hotmail.com)
Ana Armas-Padron (amarpad@gmail.com)
Pedro Ruiz-Esteban (pedro_ruiz_esteban@hotmail.com)
Armando Torres (atorresram@gmail.com)
Ricardo Gomez-Huelgas (ricardogomezhuelgas@hotmail.com)

Version: 4 Date: 1 July 2013

Author's response to reviews: see over
Dear Editor:

Thank you for the final review and comments about our manuscript #7027031399171241 entitled “Association of HbA1c and cardiovascular and renal disease in an adult Mediterranean population”.

Please find enclosed the new version of the manuscript, which includes the modification in the abstract suggested by reviewer Michal Melamed.

We hope that the manuscript is now acceptable for publication.

Yours sincerely,

M. Rosa Bernal-López
BRIEF LIST OF CHANGES MADE IN THE MANUSCRIPT

1. In the abstract, we have changed the sentence “The patients with known diabetes has a 2-fold increase in CKD or CVD” by the sentence “The patients with known diabetes had a 2-fold higher odds of CKD or CVD”.